Author’s response to reviews

Title: A cluster randomised controlled trial evaluating the effectiveness of a structured pulmonary rehabilitation education programme for improving the health status of people with chronic obstructive pulmonary disease (COPD): The PRINCE Study.

Authors:

Kathy Murphy (kathy.murphy@nuigalway.ie)
Dympna Casey (dympna.casey@nuigalway.ie)
Declan Devane (declan.devane@nuigalway.ie)
Adeline Cooney (adeline.cooney@nuigalway.ie)
Bernard McCarthy (bernard.mccarthy@nuigalway.ie)
Lorraine Mee (lorraine.mee@nuigalway.ie)
Martina Nichulain (martinanichulain@nuigalway.ie)
Andrew W Murphy (andrew.murphy@nuigalway.ie)
John Newell (john.newell@nuigalway.ie)
Eamon O’Shea (eamon.oshea@nuigalway.ie)

Version: 3 Date: 12 January 2011

Author’s response to reviews: see over
12.01.11

Dear Editor

We have now attended to all the major revisions

**Major revisions**

Abstract: the trial registration number is mentioned in the abstract, we have also updated the abstract in the submission system to ensure it remains identical to the manuscript.

References: References are cited in the text using consecutive numbers in square brackets.

Table 1: has been created using a Tables tool, does not include vertical lines

**Minor revisions**

Highlighting/tracking: we have removed all highlighting and tracking from the manuscript.

Yours Sincerely

Dympna Casey